Author’s response to reviews

Title: The accuracy of parent-reported height and weight for 6-12 year old U.S. children

Authors:

Davene Wright (davene.wright@seattlechildrens.org)
Karen Glanz (kglanz@upenn.edu)
Trina Colburn (trina.colburn@seattlechildrens.org)
Shannon Robson (robson@udel.edu)
Brian Saelens (brian.saelens@seattlechildrens.org)

Version: 2 Date: 23 Jan 2018

Author’s response to reviews:

1) Abstract Headings

Please edit the heading 'Objective' within the abstract to 'Background'. Please ensure all changes to the abstract are copied into the submission system.

This heading has been corrected

2) Author Email

Please note that the email for author Shannon M. Robson, differs between the submission system and the title page. Please edit this so the emails match.

Thank you for bringing this to our attention. This typo has been corrected in the submission system.

3) Title Page

Please remove the clinical trial registration section from the title page.

Clinical trial registration has been removed from the abstract.

4) List of Abbreviations
Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Abbreviations have been moved from the title page to just before the declarations section and all abbreviations are accounted for.

5) Consent to participate

Please clarify whether parental consent was written or verbal. If verbal please explain why and whether the ethical review board approved this method of consent.

Consent was written. This is now noted in the declarations.

6) Author Contributions

Please note author SR is missing from this section. Please ensure the contributions of all authors are included within this section.

SR’s contributions have been clarified in this section.

7) Potential Conflict of Interest

Please remove this duplicated section from the manuscript.

This duplicated section has been deleted.

8) Figure Legends

Please ensure this section is entitled 'Figure Legends'.

A title has been added to this section.

9) Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All
relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

All tracked changes have been accepted.